Author's response to reviews

Title: Potential risk of alpha-glucosidase inhibitor administration in prostate cancer external radiotherapy by the exceptional intestinal gas production: a case report

Authors:

Takuya Nishimura (t-nishi@koto.kpu-m.ac.jp)
Hideya Yamazaki (hideya10@hotmail.com)
Kazuki Iwama (k-iwama@takedahp.or.jp)
Yoshitaka Oota (oota@takedahp.or.jp)
Hironori Aibe (a-ib-n24@koto.kpu-m.ac.jp)
Satoaki Nakamura (satoaki@nakamura.pro)
Ken Yoshida (rad113@poh.osaka-med.ac.jp)
Haruumi Okabe (h-okabe@takedahp.or.jp)
Kei Yamada (kyamada@koto.kpu-m.ac.jp)

Version: 6 Date: 14 February 2014

Author's response to reviews:

February 14, 2014

Dear Editor

I am sending herewith a re-revised manuscript entitled "Potential risk of alpha-glucosidase inhibitor administration in prostate cancer external radiotherapy by the exceptional rectal gas production: a case report" to be submitted to Journal of Medical Case Reports. Your kind arrangements for publication of this paper are so much appreciated.

yours sincerely,

Hideya YAMAZAKI M.D.

Reply to Editor

1. Please include the Ethnicity of the patient in the Abstract and Case presentation sections.

We wrote “A 68-year-old mongoloids man”. We thought Japanese is a mongoloids.
We changed it to Japanese.

2. Please change the description of the patient’s gender to ‘male’.

We changed them accordingly.

3. Please reposition the Consent statement section after the Conclusions section.

We changed them accordingly.
4. Please reposition the list of abbreviations used in the manuscript and their meanings after the Conclusions section.
   We changed them accordingly.

5. Please reposition the Competing Interests section before the Reference list.
   We changed them accordingly.